Author’s response to reviews

Title: Mechanism of Worsening Diabetic Retinopathy with Rapid Lowering of Blood Glucose: The Synergistic Hypothesis

Authors:
Ahmadou Jingi (jingiahmadoumusa@yahoo.co.uk)
Aurel Tankeu (aurelnet2014@yahoo.com)
Narcisse Ateba (nateba2@yahoo.fr)
Jean Jacques Noubiap (noubiapjj@yahoo.fr)

Version: 2 Date: 23 Jun 2017

Author’s response to reviews:

TO THE EDITOR

BMC Endocrine disorders

Manuscript BEND-D-17-00094R1

Dear Editor,

We thank all the editorial team for their careful reading of our manuscript and for their helpful comments and suggestions. We have made changes accordingly and we believe that the result is a greatly improved paper. The changes in the manuscript are highlighted in Yellow. Our point-by-point replies to the editorial’s concerns is below. We are most grateful for the opportunity to publish in BMC Endocrine disorders and thank you for your consideration.

Sincerely,

The authors

1. We note that you have not included the contributions of author Narcisse Assene Ateba (NAA) in the Authors’ contributions section. Please list each individual author and their contributions in this section.

Authors: Thank you dear editor, This has been added in the contributions section
2. We also note that you have not included author Narcisse Assene Ateba in the Authors’ information section. Please include the information for all authors in this section.

Authors: Thank you dear editor. This has been revised accordingly.

3. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.

Authors: A list of abbreviations has been included before "Declarations".